Author's response to reviews

Title: WHO Systematic Review of Prevalence of Chronic Pelvic Pain: a neglected reproductive health morbidity

Authors:

Pallavi M Latthe (pallavi@doctors.org.uk)
Manish M Latthe (manish@latthe.freeserve.co.uk)
Lale Say (sayl@who.int)
Metin A Gulmezoglu (gulmezoglum@who.int)
Khalid S Khan (khalid.khan@bwhct.nhs.uk)

Version: 2 Date: 16 September 2005

Author's response to reviews:

Title revised to highlight the fact that while so much money is spent on the problem it is not on public health radar screens as an important condition